Author’s response to reviews

Title: Function of ankle ligaments for subtalar and talocrural joint stability during an inversion movement – an in vitro study

Authors:

Lu Li (mage_lilu@hotmail.com)

Albert Gollhofer (ag@sport.uni-freiburg.de)

Heinz Lohrer (lohrer@esn-ortho.de)

Nadja Dorn-Lange (dorn.lange@gmail.com)

Guiseppe Bonsignore (g.bonsignore@gmx.de)

Dominic Gehring (Dominic.Gehring@sport.uni-freiburg.de)

Version: 2 Date: 25 Feb 2019

Author’s response to reviews:

Dear Editor, dear Reviewers,

We would like to show our sincere gratitude to you, the editor and the reviewers, for the time and efforts you have invested helping us to complete our manuscript. Both reviewers’ valuable comments do contribute a lot to improve the quality of our manuscript and make it suitable for publishing. Thanks to the first reviewer for your understanding and consent with our revision. We also appreciate the second reviewer’s comment and we realize the suggestion to describe our research purpose more precisely. The sentence has been modified according to your suggestion, helping readers to have a clearer understanding of the purpose of our study.

We are looking forward to your response.